Author's response to reviews

Title: Spontaneous Atraumatic Fracture of Body of Second Cervical Vertebra: An Unusual Manifestation of Tubercular Infection

Authors:

SUMIT GUPTA Dr (drsumitlhmc@gmail.com)
SATYA PATRA Dr (drsatyarp@gmail.com)
ASMITA PARIHAR Dr (dr.asmitaparihar@gmail.com)

Version: 4 Date: 15 February 2012

Author's response to reviews: see over
Spontaneous Atraumatic Fracture of Cervical Vertebra

In Tuberculosis: A Case Report

Dr Sumit Gupta 1¹*, Dr Satya Patra 2², Dr Asmita Parihar 3³

Address: ¹Lady Hardinge Medical College
          Pocket A 45 B Dilshad Garden
          New Delhi 110095;

²Dr Ram Manohar Lohia Hospital
   New Delhi 110001;

³University College of Medical Sciences & G.T.B. hospital
   New Delhi 110092

Email: drsumitlhmc@gmail.com*
       drsatyarp@gmail.com
       dr.asmitaparihar@gmail.com